Author's response to reviews

Title: Alternative regression models to assess increase in childhood BMI

Authors:

Andreas Beyerlein (andreas.beyerlein@med.uni-muenchen.de)
Ludwig Fahrmeir (ludwig.fahrmeir@stat.uni-muenchen.de)
Ulrich Mansmann (mansmann@ibe.med.uni-muenchen.de)
André M Toschke (michael.toschke@kcl.ac.uk)

Version: 5 Date: 19 August 2008

Author's response to reviews:

- Abstract: "In contrast, smoking in pregnancy was not directly, and breastfeeding..." -> Comma after 'directly'

- Introduction: "Several risk factors for increased adult body size had different effects on specific quantiles." -> unnecessary dot deleted

- Introduction: "Comparisons between different regression models were discussed, but not quantified by model fit criteria such as Akaike Information Criterion (AIC)" -> 'Akaike Information Criterion (AIC)' inserted